UNIVERSITYOF BIRMINGHAM

University of Birmingham Research at Birmingham

Predicting seizures in pregnant women with epilepsy

Allotey, John; Fernandez-Felix, Borja M.; Zamora, Javier; Moss, Ngawai; Bagary, Manny; Kelso. Andrew ; Khan, Rehan; van der Post, Joris A. M. ; Mol, Ben; Pirie, Alexander M. ; McCorry, Dougall ; Khan, Khalid S.; Thangaratinam, Shakila

DOI: 10.1371/journal.pmed.1002802

License: Creative Commons: Attribution (CC BY)

Document Version Publisher's PDF, also known as Version of record

Citation for published version (Harvard):

Allotey, J, Fernandez-Felix, BM, Zamora, J, Moss, N, Bagary, M, Kelso, A, Khan, R, van der Post, JAM, Mol, B, Pirie, AM, McCorry, D, Khan, KS & Thangaratinam, S 2019, 'Predicting seizures in pregnant women with epilepsy: development and external validation of a prognostic model', PLoS Medicine, vol. 16, no. 5, e1002802. https://doi.org/10.1371/journal.pmed.1002802

Link to publication on Research at Birmingham portal

Publisher Rights Statement: © 2019 Allotey et al.

General rights

Unless a licence is specified above, all rights (including copyright and moral rights) in this document are retained by the authors and/or the copyright holders. The express permission of the copyright holder must be obtained for any use of this material other than for purposes permitted by law.

•Users may freely distribute the URL that is used to identify this publication.

•Users may download and/or print one copy of the publication from the University of Birmingham research portal for the purpose of private study or non-commercial research.

•User may use extracts from the document in line with the concept of 'fair dealing' under the Copyright, Designs and Patents Act 1988 (?) Users may not further distribute the material nor use it for the purposes of commercial gain.

Where a licence is displayed above, please note the terms and conditions of the licence govern your use of this document.

When citing, please reference the published version.

Take down policy

While the University of Birmingham exercises care and attention in making items available there are rare occasions when an item has been uploaded in error or has been deemed to be commercially or otherwise sensitive.

If you believe that this is the case for this document, please contact UBIRA@lists.bham.ac.uk providing details and we will remove access to the work immediately and investigate.



Citation: Allotey J, Fernandez-Felix BM, Zamora J, Moss N, Bagary M, Kelso A, et al. (2019) Predicting seizures in pregnant women with epilepsy: Development and external validation of a prognostic model. PLoS Med 16(5): e1002802. https://doi.org/10.1371/journal.pmed.1002802

Academic Editor: Lucy C Chappell, King's College London, UNITED KINGDOM

Received: November 15, 2018

Accepted: April 11, 2019

Published: May 13, 2019

Copyright: © 2019 Allotey et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Data cannot be shared publicly because it belongs to the sponsor Queen Mary University of London and consent was not sought from participants to share publicly. Data are available from the sponsor Institutional Data Access / Ethics Committee (contact via the chairperson, Sally Kerry at <u>pctu-data-</u> <u>sharing@qmul.ac.uk</u>) for researchers who meet the criteria for access to confidential data.

Funding: The EMPiRE trial was funded by a grant from the National Institute of Health Research

RESEARCH ARTICLE

Predicting seizures in pregnant women with epilepsy: Development and external validation of a prognostic model

John Allotey^{1,2}, Borja M. Fernandez-Felix^{3,4}, Javier Zamora^{1,3,4}*, Ngawai Moss⁵, Manny Bagary⁶, Andrew Kelso⁷, Rehan Khan⁸, Joris A. M. van der Post⁹, Ben W. Mol¹⁰, Alexander M. Pirie¹¹, Dougall McCorry⁷, Khalid S. Khan^{1,2}, Shakila Thangaratinam^{1,2}

1 Barts Research Centre for Women's Health, Barts and the London School of Medicine and Dentistry, Queen Mary University of London, London, United Kingdom, 2 Multidisciplinary Evidence Synthesis Hub, Queen Mary University of London, London, United Kingdom, 3 CIBER Epidemiology and Public Health, Madrid, Spain, 4 Clinical Biostatistics Unit, Hospital Ramón y Cajal, Madrid, Spain, 5 Patient and Public Involvement, Katie's Team, Katherine Twining Network, Queen Mary University of London, London, United Kingdom, 6 Queen Elizabeth Hospital Birmingham, University Hospitals Birmingham NHS Foundation Trust, Birmingham, United Kingdom, 7 Department of Neurology, Royal London Hospital, Barts Health NHS Trust, London, United Kingdom, 8 Department of Obstetrics and Gynaecology, Royal London Hospital, Barts Health NHS Trust, London, United Kingdom, 9 Department of Obstetrics and Gynaecology, University of Amsterdam, Academic Medical Centre, Amsterdam, The Netherlands, 10 Monash University, Monash Medical Centre, Clayton, Victoria, Australia, 11 NHS Education for Scotland, Edinburgh, United Kingdom

* javier.zamora@hrc.es

Abstract

Background

Seizures are the main cause of maternal death in women with epilepsy, but there are no tools for predicting seizures in pregnancy. We set out to develop and validate a prognostic model, using information collected during the antenatal booking visit, to predict seizure risk at any time in pregnancy and until 6 weeks postpartum in women with epilepsy on antiepileptic drugs.

Methods and findings

We used datasets of a prospective cohort study (EMPiRE) of 527 pregnant women with epilepsy on medication recruited from 50 hospitals in the UK (4 November 2011–17 August 2014). The model development cohort comprised 399 women whose antiepileptic drug doses were adjusted based on clinical features only; the validation cohort comprised 128 women whose drug dose adjustments were informed by serum drug levels. The outcome was epileptic (non-eclamptic) seizure captured using diary records. We fitted the model using LASSO (least absolute shrinkage and selection operator) regression, and reported the performance using C-statistic (scale 0–1, values > 0.5 show discrimination) and calibration slope (scale 0–1, values near 1 show accuracy) with 95% confidence intervals (CIs). We determined the net benefit (a weighted sum of true positive and false positive classifications) of using the model, with various probability thresholds, to aid clinicians in making individualised decisions regarding, for example, referral to tertiary care, frequency and intensity PLOS MEDICINE

(NIHR) Health Technology Assessment (HTA) Program (09/55/38). No financial support was received for the prognostic model developed and validated in this article. BWM is supported by an NHMRC Practitioner Fellowship (GNT1082548).

Competing interests: I have read the journal's policy and the authors of this manuscript have the following competing interests: AMP has been paid to provide medico-legal reports on the standard of care of women with epilepsy in pregnancy and use of anti-epileptic drugs, and has jointly held grants from the NIHR and Epilepsy Action for research on epilepsy in pregnancy. The other authors have declared that no competing interests exist.

Abbreviations: EEG, electroencephalogram; IPD, individual participant data.

of monitoring, and changes in antiepileptic medication. Seizures occurred in 183 women (46%, 183/399) in the model development cohort and in 57 women (45%, 57/128) in the validation cohort. The model included age at first seizure, baseline seizure classification, history of mental health disorder or learning difficulty, occurrence of tonic-clonic and non-tonic-clonic seizures in the 3 months before pregnancy, previous admission to hospital for seizures during pregnancy, and baseline dose of lamotrigine and levetiracetam. The C-statistic was 0.79 (95% CI 0.75, 0.84). On external validation, the model showed good performance (C-statistic 0.76, 95% CI 0.66, 0.85; calibration slope 0.93, 95% CI 0.44, 1.41) but with imprecise estimates. The EMPiRE model showed the highest net proportional benefit for predicted probability thresholds between 12% and 99%. Limitations of this study include the varied gestational ages of women at recruitment, retrospective patient recall of seizure history, potential variations in seizure classification, the small number of events in the validation cohort, and the clinical utility restricted to decision-making thresholds above 12%. The model findings may not be generalisable to low- and middle-income countries, or when information on all predictors is not available.

Conclusions

The EMPiRE model showed good performance in predicting the risk of seizures in pregnant women with epilepsy who are prescribed antiepileptic drugs. Integration of the tool within the antenatal booking visit, deployed as a simple nomogram, can help to optimise care in women with epilepsy.

Author summary

Why was this study done?

- Pregnant women with epilepsy are at increased risk of death and complications from seizures; their high-risk status during pregnancy and after childbirth is often not recognised.
- Knowledge of an individual's risk of seizures could help healthcare professionals and pregnant women make decisions regarding management.
- To our knowledge, there are currently no models to predict risk of seizures in pregnant women with epilepsy on medication.

What did the researchers do and find?

• We developed the EMPiRE model to predict the risk of seizures in pregnancy and up to 6 weeks after delivery in women with epilepsy on medication whose drug doses were managed based on clinical findings; we validated the model in a separate group of women whose dose management was based on drug levels in the blood.

- The model discriminated well between those with and without seizures, with good
 agreement between predicted and observed risks across both low- and high-risk
 women.
- The model is clinically useful for decision-making where the threshold of choice for seizure risk is between 12% and 99%.
- The model showed promising transportability to the validation cohort.

What do these findings mean?

- The EMPiRE prediction model can be used by healthcare professionals to identify pregnant women at high risk of seizures and to plan early referral for specialist input; determine the need for close monitoring in pregnancy, labour, and after childbirth; and assess antiepileptic drug management.
- The performance of the model is unlikely to vary with the antiepileptic drug dose management strategy.

Introduction

Women with epilepsy are 10 times more likely to die in pregnancy than those without the condition [1]—seizures are a common cause of death [2]. Despite warnings from consecutive reports of the Confidential Enquiry into Maternal Deaths (UK) on the failings in antenatal, intrapartum, and postnatal management of women with epilepsy, care of these women remains fragmented [3,4]. A lack of recognition of the women's high-risk status by professionals in primary and in secondary care has been highlighted consistently as the main factor behind epilepsy-related maternal deaths [2,3,5]. Furthermore, up to 4 in 10 women discontinue their antiepileptic medication in pregnancy due to concerns about the effects of drugs on the fetus, thereby increasing their risk of seizures [6,7]. Many maternal deaths in women with epilepsy could be averted with timely specialist input [5]. Seizures in pregnancy also have a negative impact on daily living. For example, the loss of driving license following seizures affects employment, relationships, and quality of life [8–10].

Pregnant women with epilepsy at risk of seizures need a personalised management plan for antenatal, intrapartum, and postnatal care, which requires multidisciplinary input through joint obstetric neurology clinics; however, these clinics are not available in all healthcare centres [11]. Furthermore, women at high risk of seizures need close monitoring in labour, with adequate pain relief measures such as epidural analgesia, and use of long-acting benzodiaze-pines such as clobazam [11]. Current guidelines recommend the use of these measures in high-risk women [11]. But a lack of guidance on what constitutes high-risk pregnancy is one factor that has contributed to variations in the care of pregnant women with epilepsy [11].

Prediction of seizures based on a woman's individual characteristics not only provides an accurate picture of the risks to inform decision-making, but also promotes effective communication between the multi-specialty teams caring for women with epilepsy. A tool for predicting seizure risk can empower women to make informed decisions on their antenatal and intrapartum care. Furthermore, awareness of one's risk status may lower any anxiety arising from the

unpredictable nature of seizures [12], and promote adherence to medication through risk-informed counselling [13].

To our knowledge, there are currently no models to predict seizure risk in pregnant women with epilepsy. Existing, small retrospective studies provide imprecise estimates of the performance of individual predictors, such as type of seizures and seizure status in pre-pregnancy [14–16]. We aimed to develop and externally validate a prognostic model to predict the risk of seizures in pregnant women with epilepsy on medication, until 6 weeks postpartum. We also planned to determine the net benefit of using the model at various threshold probabilities using decision curve analysis.

Methods

We developed and validated the prognostic model for seizures in the prospective multicentre EMPiRE (AntiEpileptic drug Monitoring in PREgnancy) study, which recruited pregnant women with epilepsy on antiepileptic drugs at first antenatal visit from 50 maternity units in the UK between 4 November 2011 and 17 August 2014 [17]. The UK National Research Ethics Committee approved the EMPiRE study (11/WM/0164), written consent was obtained from participants, and the protocol can be accessed at https://www.journalslibrary.nihr.ac.uk/ programmes/hta/095538#/. The research reported here did not require further review by an ethics committee. We reported our prognostic study in line with the TRIPOD (Transparent Reporting of a multivariable prediction model for Individual Prognosis Or Diagnosis) recommendations, and present our findings as a nomogram, a graphical representation of the model to calculate an individual's risk of seizure [18–20] (S1 TRIPOD Checklist).

Model development and validation cohorts

EMPiRE was a prospective study, and recruited pregnant women with epilepsy on lamotrigine, carbamazepine, phenytoin, or levetiracetam before 24 weeks' gestation. Serum antiepileptic drug levels were assessed every month, but the women and clinicians were blinded to these levels. For women for whom drug levels remained stable, the blinding was maintained (non-randomised cohort) until delivery, and drug doses were adjusted based on clinical features, in line with national recommendations [17,21]. Women whose serum drug levels fell were randomly allocated either to a strategy of adjusting antiepileptic doses based on serum drug levels (after unblinding) or to a strategy of changing the drug doses based on only clinical features (blinding maintained). All participants were followed up until 6 weeks after delivery (S1 Appendix). The study is described in detail elsewhere [17]. For model development, we used the cohort of women who were managed without routine serum drug level monitoring (non-randomised and randomised women), which is in line with standard epilepsy care in the UK [17,21]. We validated the model in the separate cohort of women managed differently, with routine therapeutic drug monitoring, as practised in some countries such as the US, to determine if the model was transportable across varied healthcare practices [22].

Candidate predictors

A multidisciplinary team of neurologists, obstetricians, and researchers selected the candidate predictors for further evaluation in the prognostic model, based on existing evidence and their relevance to clinical care [14,16,23–26]. From an initial list of 65 baseline variables, we selected the following candidate predictors: age at first seizure, history of learning difficulty or mental health disorder, baseline seizure classification (tonic-clonic, non-tonic-clonic, unspecified), history of seizure in the 3 months before pregnancy (tonic-clonic, non-tonic-clonic), number of seizures between start of pregnancy and baseline visit, type of antiepileptic drug taken at

baseline, dose of antiepileptic drug taken at baseline, gestational age at baseline, and hospital admission for seizures in a previous pregnancy. All continuous predictors were assumed to be linearly associated with the outcome.

Outcome

Our main outcome was the occurrence of tonic-clonic (convulsive) or non-tonic-clonic (nonconvulsive) seizure [27]. Participants prospectively recorded their epileptic seizures, if any, in purpose-built seizure diaries. To avoid overfitting of multivariable models, the rule of thumb is to ensure that there are 10 events for each predictor variable that was considered for inclusion in the model [28]; we worked within this rule limitation.

We assessed the predictive performance of the model using measures of discrimination (C-statistic) and accuracy (calibration slope). The C-statistic represents the ability of the model to discriminate between those who do and do not experience seizures; a value of 1 indicates perfect discrimination, and a value of 0.5 indicates no discrimination beyond chance [29]. Models are considered to have a good performance when the C-statistic exceeds 0.7 [30]. Calibration refers to agreement between the predicted and observed risk of seizure for all groups of predicted probabilities. A well-calibrated model will have a calibration slope of 1, and all groups will fit close to this line.

Statistical analysis

Model development. Missing values were imputed using 10-fold multiple imputation by chained equations. Within each imputed dataset we developed a multivariable logistic regression model using the LASSO (least absolute shrinkage and selection operator) method, which simultaneously selects the variables and penalises the model coefficients for over-optimism. We selected the lambda parameter that minimised expected model deviance. Final coefficients were combined across imputed datasets using Rubin's rule [31,32]. Confidence intervals for model coefficients were obtained by bootstrap sampling. Bootstrap validation was carried out to adjust the performance of the model for optimism. We repeated the entire modelling process on 100 bootstrap samples drawn from each of the 10 imputed datasets. We did not consider the number of tonic-clonic seizures between the diagnosis of pregnancy and baseline visit as a variable, as it was highly correlated with the predictor variable of tonic-clonic seizure history in the 3 months before pregnancy.

We assessed the overall discriminatory ability of the model using the C-statistic (summarised as the area under receiver operating characteristic curve [AUC ROC]) with 95% confidence interval (CI). Model calibration was visually assessed with a calibration plot representing deciles of predicted probability of seizure against the observed rate in each risk group. The calibration slope was also calculated, which is the slope of the regression line fitted between predicted and observed risk probabilities on the logit scale, with 1 being the ideal value.

External validation. We externally validated the final model in a separate cohort of women whose drug levels were routinely monitored (therapeutic drug monitoring) to plan drug dose adjustments. We report the predictive performance in this cohort using the same measures of discrimination and calibration as used in the model development cohort [33]. We visually assessed the model's calibration by plotting quartiles of predicted probability of seizure against the observed rate in each quartile, and estimated the calibration slope [34,35].

Sensitivity analysis. We undertook sensitivity analysis by combining all available data in the development and validation cohorts to determine if there was any change in model

performance. We evaluated antiepileptic drug dose monitoring strategy (therapeutic drug monitoring versus clinical features monitoring) as a predictor while developing the combined model.

Decision curve analysis. We performed decision curve analysis to assess the clinical value of the model. A decision rule should take into consideration the identification of women likely to have seizures because of the significant consequences (maternal death, accidents, loss of driving license), and the avoidance of unnecessary interventions leading to adverse impact on mother or baby. We determined the net benefit of the model across a wide range of threshold probabilities, instead of simply classifying all pregnant women with epilepsy as predicted to have seizures or classifying no women as predicted to have seizures [36]. We represent the net benefit as a function of the decision threshold in a decision curve plot.

The choice of thresholds will vary according to the planned intervention, the preference of the clinician, and the preference of the mother. For example, if the intervention involves referral of a pregnant woman with epilepsy to a tertiary unit for antenatal and intrapartum care, at a low threshold, false negatives are minimised at the expense of unnecessary referrals to tertiary care. At a high threshold, fewer women are referred, but women who are likely to benefit may be denied access to tertiary care. We expect an intermediate range of thresholds to be clinically acceptable. But if the planned intervention is to increase the dose and number of antiepileptic drugs—which can have adverse side effects on the mother and increase the risk of long-term neurodevelopmental problems in the child—clinicians may choose a higher threshold than what was chosen for tertiary referral.

Nomogram. We also developed a simple, easy-to-use nomogram to calculate the predicted probability of seizures in pregnant women at the time of antenatal booking.

The analyses were performed using Stata software version 15.1 and R software version 3.3.2 [37,38].

Results

The EMPiRE study recruited 560 pregnant women. The model development cohort included 399 women; the validation cohort included 128 women (S1 Appendix).

Characteristics of the women

The average gestational age at baseline was 16.6 weeks (SD 4.0) in the development cohort and 14.9 weeks (SD 4.4) in the validation cohort (Table 1). The mean age at first seizure was similar in both cohorts, at 16 years, and 10%–15% of women had a learning difficulty or mental health disorder. A similar proportion of women in the development and validation cohorts were previously diagnosed to have tonic-clonic seizures (development, 39%; validation, 36%). Overall, 46% (182/399) of women in the development cohort and 39% (50/128) in the validation cohort had experienced seizures in the 3 months prior to pregnancy. Lamotrigine was the commonest antiepileptic drug prescribed in both cohorts; more than half of the women took lamotrigine in the development (226/399, 57%) and validation cohorts (80/128, 63%).

Model performance

Overall, 46% (183/399) of women in the development cohort experienced 1 or more seizures at any time from baseline until 6 weeks after delivery. Tonic-clonic seizures accounted for half of all seizures (90/183, 49%). Eight predictors were significantly associated with seizures and were included in the final multivariable model: age at first seizure, history of mental health disorder or learning difficulty, baseline seizure classification (tonic-clonic, non-tonic-clonic, unspecified), hospital admission for seizures in a previous pregnancy, tonic-clonic seizure in

	MEDICINE
--	----------

Characteristic	Development coho	ort $(n = 399)$	Validation cohort (n = 128)		
	Mean (SD) or <i>n</i> (%)	Number with missing data, n (%)	Mean (SD) or <i>n</i> (%)	Number with missing data, <i>n</i> (%)	
Gestational age at baseline (weeks)	16.6 (4.0)	0	14.9 (4.4)	0	
History of learning difficulty or mental illness	50 (13%)	1 (0.3%)	20 (16%)	0	
Age at first seizure (years)	16.5 (7.4)	5 (1.3%)	16.8 (7.8)	0	
≤ 10 years	70 (17.8%)		28 (21.9%)		
11–20 years	215 (54.6%)		59 (46.1%)		
21-30 years	94 (23.9%)		35 (27.3%)		
31-40 years	15 (3.8%)		6 (4.7%)		
Admission to hospital for seizures in previous pregnancy	28 (7.0%)	22 (5.5%)	14 (10.9%)	7 (5.5%)	
Seizure classification at baseline					
Tonic-clonic	155 (39%)	0	46 (36%)	0	
Non-tonic-clonic	232 (58%)	0	78 (61%)	0	
Unspecified	12 (3%)	0	4 (3%)	0	
Seizure in the 3 months before pregnancy	182 (46%)		50 (39%)		
Tonic-clonic	52 (13%)	83 (20.8%)	12 (9%)	32 (25.0%)	
Non-tonic-clonic	130 (33%)	33%) 0 38 (3		0	
Number of seizures in pregnancy prior to the baseline visit					
Tonic-clonic	0.7 (2.8)	82 (20.6%)	0.4 (2.1)	31 (24.2%)	
Non-tonic-clonic	11.6 (108.4)	0	18.1 (83.4)	0	
Antiepileptic drug intake at baseline					
Carbamazepine	74 (19%)	0	16 (13%)	0	
Lamotrigine	200 (50%)	0	66 (52%)	0	
Levetiracetam	99 (25%)	0	31 (24%)	0	
Phenytoin	0	0	1 (1%)	0	
Lamotrigine and carbamazepine	1 (0.3%)	0	_	0	
Lamotrigine and levetiracetam	25 (6%)	0	14 (11%)	0	
Baseline dose of antiepileptic drugs (mg/day)					
Carbamazepine	706.0 (348.5)	0	612.5 (346.2)	0	
Lamotrigine	272.1 (155.6)	0	269.4 (160.6)	0	
Levetiracetam	1,641.3 (886.8)	0	1,533.3 (760.5)	0	
Phenytoin	0	0	200 (—)	0	

Table 1. Details of women's characteristics in the development and validation cohorts of the EMPiRE prediction model and the proportion with missing data.

https://doi.org/10.1371/journal.pmed.1002802.t001

the 3 months before pregnancy, non-tonic-clonic seizure in the 3 months before pregnancy, baseline dose of lamotrigine, and baseline dose of levetiracetam (Table 2). The model is presented as a graphical calculator (nomogram) in Fig 1.

The equation of the EMPiRE prediction model for risk of seizures during pregnancy and until 6 weeks after delivery in women with epilepsy on antiepileptic drugs was as follows:

probability(seizure) = $\exp(Y)/(1 + \exp(Y))$

where Y = -1.39 + (-0.02 * age at first seizure) + 0.61 [unspecified seizures] + 0.75 [non-tonicclonic seizures] + (0.02 * dose of levetiracetam/100) + (0.29 * dose of lamotrigine/100) + 0.66 [non-tonic-clonic seizures in the 3 months before pregnancy] + 1.97 [tonic-clonic seizures in the 3 months before pregnancy] + 0.67 [learning difficulty or mental health disorder] + 0.17 [admitted to hospital for seizures during previous pregnancy].

Candidate predictor	Multivariat (<i>n</i> = 399)	Multivariable analysis after MI (<i>n</i> = 399)			
	OR	Bootstrap 95% CI			
Age at first seizure (years)	0.98	0.97, 0.99			
History of learning difficulty or mental illness	1.96	1.68, 2.89			
Seizure classification at baseline (ref. tonic-clonic)					
Non-tonic-clonic	2.11	1.88, 2.62			
Unspecified	1.85	1.64, 4.30			
Tonic-clonic seizure in the 3 months prior to pregnancy	7.20	6.63, 11.93			
Non-tonic-clonic seizure in the 3 months prior to pregnancy	1.94	1.71, 2.38			
Baseline dose of lamotrigine (×100 mg/day)	1.34	1.30, 1.44			
Baseline dose of levetiracetam (×100 mg/day)	1.02	1.01, 1.03			
Admitted to hospital for seizures in previous pregnancy	1.19	1.08, 1.92			
Baseline dose of carbamazepine (×100 mg/day)	_	_			
Number of non-tonic-clonic seizures since the start of pregnancy		_			
Gestational age at baseline (weeks)	_	_			

Table 2. Multivariable LASSO logistic regression of seizure risk prediction in pregnant women with epilepsy.

95% CI: Bootstrap limits of the confidence interval obtained from percentiles 2.5 and 97.5. Missing values were imputed using 10-fold MI by chained equations (step 1). We fitted a regression model using the LASSO strategy in each of the 10 imputed datasets (step 2). We averaged model coefficients using Rubin's rule to get the final model coefficients (step 3). To obtain non-parametric 95% confidence intervals for model coefficients, we repeated the previous step 2 and step 3 on 1,000 bootstrap samples. Limits of the 95% confidence interval for each coefficient were the 2.5th and 97.5th percentiles of their distribution.

MI, multiple imputation; OR, odds ratio.

https://doi.org/10.1371/journal.pmed.1002802.t002

All variables were coded as binary (1 when present and 0 when absent) except for age at first seizure (years), dose of lamotrigine (mg/day), and dose of levetiracetam (mg/day).

The apparent C-statistic for the model was 0.80 (95% CI 0.76, 0.85). After bootstrap adjustment for optimism, the final prediction model had a C-statistic of 0.79 (95% CI 0.75, 0.84) to discriminate between women with and without seizures (Table 3). The optimism-adjusted calibration plot (Fig 2) showed mostly good agreement between the predicted and observed risks, and the calibration slope was 1.26 (95% CI 0.98, 1.54).

Our sensitivity analysis, which combined all available data (n = 527), resulted in a model with the same predictors and similar coefficients as the EMPiRE model developed using only the development cohort. The antiepileptic drug monitoring strategy was not found to be a significant predictor of seizures and was therefore not selected in the combined model. The C-statistic and calibration slope of the combined model were 0.78 (95% CI 0.74, 0.82) and 1.22 (95% CI 0.44, 1.46), respectively (S2 Appendix).

External validation and predictive performance

In the external validation cohort, 45% (57/128) of women experienced seizures at any time from baseline until 6 weeks after delivery; tonic-clonic seizures were reported in 39% (22/57) of women who had seizures. The final model showed good discrimination when externally validated, with a C-statistic of 0.76 (95% CI 0.66, 0.85). The model showed mostly good agreement between the predicted and observed risks, with a calibration slope of 0.93 (95% CI 0.44, 1.41) (Fig 2).

	EMPIRE NOM	IOGRAM										
	Patient name											
1	Points 0	1	2	3	4	5	6	7	8	9	10	¹¹ Points
	Age at first seizure 4	0 35 30 25	20 15 10 5	5 0								
	Seizure diagnosis classification at		Unspecifi	ed								
ABLES	booking To	nic Clonic	N	on-Tonic Clo	nic							
	Levetiracetam at booking (mg/day)	0 1000 200	0 3000 4000									
	Dose of	I										
VARI	booking (mg/day)	0 100	200	300	400	500	600	700	800	900	1000	
	seizures 3 months	0	 Ye	S								
	Tonic clonic seizures											
	pregnancy N	0						Yes				
	for seizures in N	o Yes										
	Learning difficulty or ^L	10	 Ye	s								
2	Total score											<u> </u>
3	Probability of		5	10	00	15	05	20		25		30 —
	Instructions	20 31	0 40 50	60 70	80	90	95		99.9			
	Calculate individua 1. For each varia	l risk in three able on the le	steps: ft, read off cor	responding p	oints above	2						
	2. Add up the po 3. Determine th	oints to get a e correspondi	total score ing probability	of seizure								
F		0000414										
	Patient name	IOGRAM										
1	Points		2	3	4	5	6	7	8	9	10	
1	Points Age at first seizure	0 35 30 25	20 15 10	3	4	5	6	7	8	9	10	¹¹ Points
1	Points Age at first seizure Seizure diagnosis classification at	0 35 30 25	2 20 15 10 5 Unspecifi	3 5 0 ed	4	5	6	7	8	9	10	¹¹ Points
1	Points Age at first seizure Seizure diagnosis classification at booking Dose of	1 0 35 30 25 nic Cionic	2 20 15 10 s Unspecifi	3 0 ed on-Tonic Clo	4 nic	5	6	7	8	9	10	¹¹ Points 1 2.6
1	Points Age at first seizure Seizure diagnosis classification at booking TO Dose of Levetiracetam at booking (rg/day)	0 35 30 25 nic Cionic	2 20 15 10 5 Unspecifi 0 3000 4000	3 0 ed	nic	5	6	7	8	9	10	¹¹ Points 1 2.6 1.5
ABLES 1	Points Age at first seizure Seizure diagnosis classification at booking Dose of Levetiracetam at booking (mg/day) Dose of Lamotrioine at	0 35 30 25 nic Cionic 0 1000 200	20 15 10 5 Unspecifi 0 3000 4000	3 0 ed on-Tonic Clo	nic	5	6	7	8	9	10	¹¹ Points 1 2.6 1.5 4
VARIABLES 1	Points Age at first seizure Seizure diagnosis classification at booking TO Dose of Levetiracetam at booking (mg/day) Dose of Lamotrigine at booking (mg/day)	0 35 30 25 nic Conic 0 1000 200 0 1000 100	2 20 15 10 Unspecifi 0 3000 4000 	3 0 ed on-Tonic Clo	4 nic	5	6	7	8	9	10	11 Points 1 2.6 1.5 4
VARIABLES	Points Age at first seizure Classification at booking Dose of Levetiracetam at booking (mg/day) Dose of Lamotrigine at booking (mg/day) Non-Tonic clonic seizures 3 months pre-pregnancy	0 35 30 25 nic Cionic 0 1000 2000 0 100 0 0	2015 10 4 Unspecifi 0 3000 4000 0 2000 Ye	s	4 nic	5	600	7	8	9	10 10 1000	¹¹ Points 1 2.6 1.5 4 2.5
VARIABLES	Points Age at first seizure Seizure diagnosis classification at booking To Dose of Levetiracetam at booking (mg/day) Dose of Lamotrigine at booking (mg/day) Non-Tonic clonic seizures 3 months pre-pregnancy Tonic clonic seizures 3 months pre-	0 35 30 25 nic Conic 0 1000 200 0 1000 200	2 20 15 10 9 0 0 0 200 200 200 200	3 0 ed jon-Tonic Clo 300	nic 400	5	6	7	8 8 800	9		¹¹ Points 1 2.6 1.5 4 2.5
1 VARIABLES	Points Age at first seizure Classification at booking To Dose of Levetiracetam at booking (mg/day) Dose of Lamotrigine at booking (mg/day) Non-Tonic clonic seizures 3 months pre-pregnancy Tonic clonic seizures 3 months pre- pregnancy Admitted to hospital	0 35 30 25 nic Conic 0 1000 2000 0 100 0 0	2 20 15 10 5 Unspecifi 0 3000 4000 0 200 Ye	3 0 ed	400	5	600	7 700 Yes	8	9		¹¹ Points 1 2.6 1.5 4 2.5 0 0.6
1 VARIABLES	Points Age at first seizure Seizure diagnosis classification at booking To Dose of Levetiracetam at booking (mg/day) Dose of Lamotrigine at booking (mg/day) Non-Tonic clonic seizures 3 months pre-pregnancy Tonic clonic seizures 3 months pre- pregnancy Admitted to hospital for seizures in previous pregnancy	0 35 30 25 nic Clonic 0 1000 200 0 1000 200 0 0 0 1000 200 0 0 0 Yes	2 20 15 10 5 Unspecifi 0 3000 4000 200 Ye	3 on-Tonic Clo 300	400	500	600	7 700 Yes	8	9		¹¹ Points 1 2.6 1.5 4 2.5 0 0.6
1 VARIABLES 1	Points Age at first seizure Seizure diagnosis classification at booking To Dose of Levetiracetam at booking (mg/day) Dose of Lamotrigine at booking (mg/day) Non-Tonic clonic seizures 3 months pre-pregnancy Tonic clonic seizures 3 months pre- pregnancy Admitted to hospital for seizures in previous pregnancy Learning difficulty or mental health disorder	0 35 30 25 nic Conic 0 1000 2000 0 100 0 100 0 100 0 40 100 100 100 100 100 100 100 100 100 1	2 20 15 10 5 Unspectfi 0 3000 4000 0 200 Ye	s	400	5	600	7 700 Yes	8	9	10	¹¹ Points 1 2.6 1.5 4 2.5 0 0.6 0
1 AARIABLES 2	Points Age at first seizure Seizure diagnosis classification at booking Dose of Levetiracetam at booking (mg/day) Dose of Lamotrigine at booking (mg/day) Non-Tonic clonic seizures 3 months pre- pregnancy Admitted to hospital for seizures in N previous pregnancy Learning difficulty or mental health disorder Total score 0	0 35 30 25 nic Conic 0 1000 200 0 1000 200 0 0 0 1000 200 0 0 0 1000 200 0 0 0 1000 200 0 0 0 1000 200 0 1000 2000 2	2 20 15 10 5 Unspecifi 0 3000 4000 0 200 Ye	s s	400	500	600	7 700 Yes	800	9 900		¹¹ Points 1 2.6 1.5 4 2.5 0 0.6 0 0.6 0
1 VARIABLES 2 3	Points Age at first seizure Seizure diagnosis classification at booking TO Dose of Levetiracetam at booking (mg/day) Dose of Lamotrigine at booking (mg/day) Non-Tonic clonic seizures 3 months pre-pregnancy Admitted to hospital for seizures in previous pregnancy Admitted to hospital for seizures in previous pregnancy mental health disorder Total score	0 35 30 25 nic Conic 0 1000 2000 0 100 0 100 0 0 0 Yes 10 10 0 Yes	2 20 15 10 5 Unspecifi 0 3000 4000 0 200 Ye 5 5 0 40 50	s s 10 60 70	400	500	600 600	7 700 Yes	800	- 900		¹¹ Points 1 2.6 1.5 4 2.5 0 0.6 0 0 12.2
1 AARIABLES 2 3	Points Age at first seizure Seizure diagnosis classification at booking To Dose of Levetiracetam at booking (mg/day) Dose of Lamotrigine at booking (mg/day) Non-Tonic clonic seizures 3 months pre-pregnancy Admitted to hospital for seizures in N previous pregnancy Learning difficulty or mental health disorder Total score 0 Probability of seizure (%) 10 Example calculation: was on 300mm of healthouse	0 35 30 25 nic Conic 0 1000 200 0 1000 100 0 100 0 100 0 20 0 Yes 10 1 1 1 100 0 20 0 20 0 20 0 20 0 20 0	2 20 15 10 1 Unspecifi 0 3000 4000 0 200 7 Ye 5 5 0 40 50 0 man at book	s s	400	5 500 500 15 90 2ure at age 2	6 600 600 955 (1 point	7 700 700 Yes	8 800 99.9 sclassified as the	9 9 900	10 10 1000	¹¹ Points 1 2.6 1.5 4 2.5 0 0.6 0.6 0 0.6 0 12.2 30
1 ARRIABLES 3	Points Age at first seizure Seizure diagnosis classification at booking TO Dose of Levetiracetam at booking (mg/day) Dose of Lamotrigine at booking (mg/day) Non-Tonic clonic seizures 3 months pre-pregnancy Admitted to hospital for seizures in previous pregnancy Learning difficulty or Probability of seizure (%) 10 Example calculation: was on 3000mg of lev seizures in the 3 mon mental health disorder	0 35 30 25 nic Clonic 0 1000 200 0 100 100 0 100 0 100 0 Yes 0 Yes 10 1 10 0 100 100 100 100 100 100 100 1	2 20 15 10 1 Unspecifi 0 3000 4000 0 2000 Ye Ye 5 0 40 50 roman at book 5 points), and r pregnancy (2 spents), and r pregnancy (2 ut has been H	s s s s s s s s s s s s s s s s s s s	nic 400 400 her first sei notrigine (t no tonic-t a result do	5 500 500 zure at age 2 4 points) wh clonic seizur seizures to	6 600 600 95 25 (1 point ten she wa se in this ti her previo	7 700 Yes), which wa s seen ante ime (0 point us preprance	8 800 99.9 s classified as natally at boc s). She has na y (0.6 points	9 9 900 900	10 100 1000	¹¹ Points 1 2.6 1.5 4 2.5 4 0 0.6 0 0 12.2 30 12.2

Fig 1. EMPiRE nomogram for predicting the risk of seizures at antenatal booking in pregnant women with epilepsy on antiepileptic drugs: A worked example.

https://doi.org/10.1371/journal.pmed.1002802.g001

Net benefit of model use

In our decision curve analysis (Fig.3), the curve for the EMPiRE model showed positive net benefit for predicted probability thresholds between 12% and 99% compared to managing

Table 3. EMPiRE model performance.

Performance measure	Development cohort (<i>n</i> = 399)	Validation cohort ($n = 128$)
C-statistic	0.79 (95% CI 0.75, 0.84)	0.76 (95% CI 0.66, 0.85)
Calibration slope	1.26 (95% CI 0.98, 1.54)	0.93 (95% CI 0.44, 1.41)

https://doi.org/10.1371/journal.pmed.1002802.t003



Fig 2. Calibration of the EMPiRE prediction model by comparing observed versus predicted risk of seizures in pregnant women on antiepileptic drugs, with a frequency histogram. Top panel = development cohort; bottom panel = validation cohort.

https://doi.org/10.1371/journal.pmed.1002802.g002

pregnant women with epilepsy as if they will all have seizures or managing them as if none of them will have seizures (i.e., treat-all or treat-none strategies). <u>Table 4</u> provides estimates of the net benefit of using the model for various probability thresholds. For low thresholds, below 12%, there was no difference between using the EMPiRE model and treating women as if they will all have seizures.

Discussion

Summary of the findings

The EMPiRE model performs well in predicting the risk of seizures at the time of antenatal booking in pregnant women with epilepsy who are prescribed antiepileptic medication. The



Fig 3. Decision curve analysis using the EMPiRE seizure risk prediction model. Red line (treat none) = net benefit when we assume that no pregnant woman with epilepsy will have the outcome (seizure in pregnancy); blue line (treat all) = net benefit when we assume that all pregnant women with epilepsy will have the outcome; green line (EMPiRE nomogram) = net benefit when we manage pregnant women with epilepsy according to the predicted risk of the outcome (seizure in pregnancy) estimated by the EMPiRE model. The preferred strategy is the one with the highest net benefit at any given threshold.

https://doi.org/10.1371/journal.pmed.1002802.g003

model incorporates routinely available characteristics that are easy to measure, such as age at first seizure, type of seizures, seizures in the 3 months before pregnancy, mental health, admission to hospital for seizures during a previous pregnancy, and dose of antiepileptic drugs. The model is clinically useful over a range of threshold probabilities, and is relevant to general practitioners, epilepsy specialists, obstetricians, and midwives in identifying high-risk women. The model shows potential for transportability across risk groups to settings where routine therapeutic drug monitoring is undertaken, but the findings should be interpreted with caution due to the small number of events in the validation sample. Our simple nomogram is designed to facilitate the model's use in clinical practice.

Strengths and limitations

To our knowledge, ours is the only clinical prognostic model to predict seizures in pregnant women with epilepsy. We developed the model using data from a prospective, high-quality, multicentre study. We evaluated predictors that were clinically relevant and routinely available to healthcare professionals, so that the model can be easily applied in clinical practice. Missing values of predictors were dealt with by multiple imputation, thereby avoiding loss of useful information [39,40]. We developed the model to predict seizures not only in pregnancy, but up to 6 weeks after delivery, a period with increased risks to the mother and baby [11,41]. We adjusted for optimism and addressed issues around overfitting in the model. In addition to providing the model as an easy-to-use nomogram, we provided information on its clinical use at various threshold probabilities for decision-making [35]. The model includes clinical

Threshold probability	Net benefit		Advantage of using the model			
	Treat all women	EMPiRE model	Difference in net benefit	Reduction in number who do not need the intervention per 100 women		
0.05	0.430	0.430	0	0		
0.1	0.398	0.398	0	0		
0.15	0.363	0.368	0.003	2		
0.2	0.323	0.329	0.008	3		
0.25	0.278	0.299	0.018	6		
0.3	0.227	0.276	0.049	11		
0.35	0.167	0.266	0.100	19		
0.4	0.098	0.248	0.150	22		
0.45	0.016	0.213	0.207	25		
0.5	-0.083	0.193	0.286	29		
0.55	-0.203	0.166	0.372	30		
0.6	-0.353	0.164	0.503	34		
0.65	-0.547	0.120	0.663	36		
0.7	-0.805	0.111	0.911	39		
0.75	-1.165	0.088	1.253	42		
0.8	-1.707	0.073	1.774	44		
0.85	-2.609	0.057	2.668	47		
0.9	-4.414	0.030	4.454	49		
0.95	-9.827	0.023	9.852	52		
0.99	-53.135	0.000	53.135	54		

Table 4. Net benefit of using the EMPiRE prediction model compared to managing women with epilepsy assuming all of them will have seizures in pregnancy or the postpartum period.

https://doi.org/10.1371/journal.pmed.1002802.t004

variables that are easily accessible at the time of booking for incorporation into an app or integration into computer systems within healthcare services. A convenient and easy-to-use nomogram of the EMPiRE model allows for immediate use of the model to predict the risk of seizures without the need to remember the formulae behind it.

Our model development and validation approach took into account the significant variations in the management of antiepileptic drug dosages in pregnancy to prevent seizures [42]. While the American Academy of Neurology recommends routine serum therapeutic drug monitoring, with dosage increased if the serum drug level falls [22], the UK National Institute for Health and Care Excellence, Royal College of Obstetricians and Gynaecologists, and Scottish Intercollegiate Guidelines Network guidelines do not recommend routine drug monitoring but drug dose adjustments based mainly on clinical features [11,21,43]. In this paper, we developed and internally validated the model in women whose drug dose was managed based on clinical features to determine the accuracy and reproducibility of the model. Through our external validation, we assessed the transportability of the model to women managed using a different management strategy (therapeutic drug monitoring), which appears promising [44]. Furthermore, when we developed the combined model by using all available data (including women routinely and not routinely monitored for drug levels) in our sensitivity analysis, we did not observe any differences either in the number and type of predictors or in the model's performance compared to the model developed using only women in the development cohort. The antiepileptic drug dose monitoring strategy was not identified to be a significant predictor in the combined model, implying that the model is generalisable irrespective of the strategy.

There are some limitations to this study. The cohorts consisted of women recruited with pre-specified criteria, which may limit the use of the model in all women [19,45]. We did not

include women on sodium valproate; this is consistent with current recommendations against valproate use in pregnancy, due to the increased risk of birth defects and neurodevelopmental disorders [46]. The model can only be used in women managed on phenytoin, lamotrigine, levetiracetam, or carbamazepine and when information is available on all predictors in the context of care in a high-income setting. This limits its transportability to low- and middleincome countries with resource constraints and non-availability of these drugs [47]. Women were recruited at varied gestational ages. However, we evaluated gestational age as a predictor, and it was not selected by the modelling strategy in the final model. The model included history of seizures in the 3 months before pregnancy obtained through retrospective recall, with resultant bias. We consider this to reflect the real life scenario, where women who do not receive pre-pregnancy specialist epilepsy care often do not maintain a prospective seizure diary prior to seeing an epilepsy specialist in pregnancy. It is possible that a different predictor such as history of seizures in the 9 months before pregnancy instead of the 3-month history in our model may have improved its performance [48,49]. We only included clinical predictors routinely available at the time of antenatal booking, and did not evaluate other tests such as electroencephalogram (EEG) or MRI, or risk factors such as history of nocturnal or prolonged seizures, which may be available in specialist epilepsy care. We could not assess any changes in antiepileptic medication before conception because this information was not routinely recorded at antenatal booking, and was not collected in the EMPiRE trial. These additional variables may have improved the performance of the model. Due to the small sample size and the small number of events in the validation cohort (<100), we were limited in our interpretation of the transportability of the model [50].

Comparison to existing evidence

To our knowledge, 2 other prediction models exist for seizures, both involving non-pregnant individuals: seizure prediction in children and adults who have recently stopped their antiepileptic drugs, reported using individual participant data (IPD) meta-analysis, and prediction of subsequent seizures after a single seizure in individuals without clear indication to commence treatment (MESS study) [51,52]. Some predictors in the IPD meta-analysis model such as the age at onset of epilepsy were also present in our model. It is not appropriate or feasible to apply other variables, such as seizure-free interval before antiepileptic drug withdrawal and epileptiform abnormality on EEG, to the pregnant population [51]. The performance of our EMPiRE model was better than that of the IPD meta-analysis model (C-statistic 0.65) [51]. The MESS study, which used split sample validation, did not report the performance of the model with currently recommended measures such as C-statistic and calibration slope, and hence we are unable to compare that model with the EMPiRE model.

Other individual studies such as the EURAP (European and International Registry of Antiepileptic Drugs and Pregnancy) have reported on the association between maternal risk factors and seizures in pregnancy—none provided multivariable prognostic models [14,16]. Compared to a third of pregnant women with epilepsy on medication developing seizures in the EURAP study, 46% of women in the EMPiRE cohorts experienced seizures in pregnancy or until 6 weeks after delivery [14,23]. This difference could be attributable to the known increase in seizures that occurs after delivery in new mothers, as EURAP did not include postnatal mothers or prospective seizure diaries [41]. Inclusion of a selective group of women in the EMPiRE study may also have contributed to the difference. Similarly to the EURAP study, our final model identified lamotrigine dose to be a predictor of seizures [14]. Another small retrospective study identified pre-pregnancy seizure status to be the main predictor of seizures in pregnancy [16], which was also the strongest predictor of seizures in our model.

Relevance to clinical care

Currently pregnant women with epilepsy are managed by varied healthcare professionals such as general practitioners, obstetricians and midwives, and epilepsy specialists. There is no clear pathway for multidisciplinary communication. Joint obstetric neurology clinics are not available in half of the maternity units in UK, a major hindrance for integration of epilepsy care within antenatal care [53]. The first step towards achieving integrated care is effective risk communication of the mother's seizure status. Such a risk-based approach using quantified risk estimates can help to avoid maternal deaths such as those reported in the MBRRACE-UK (Mothers and Babies: Reducing Risk through Audits and Confidential Enquiries across the UK) report [5], where women with epilepsy were never treated by an epilepsy specialist in pregnancy, and were left unmonitored during their hospital admissions without specialist input [5].

We refrained from recommending specific decision thresholds for various interventions as these are likely to vary with the potential adverse effects and costs of the planned intervention. For example, primary care clinicians may consider a 20% cutoff, a level of risk associated with driving restrictions, to be appropriate to make decisions on early referral to tertiary units with joint obstetric neurology clinics. However, secondary care clinicians may choose a higher threshold when the intervention involves frequent antenatal monitoring (weekly or fortnightly), intrapartum use of invasive interventions for pain relief such as epidural and other medications (such as clobazam, which carries a risk of neonatal respiratory depression), or close monitoring in the postnatal period. The choice of threshold in a clinical setting is also likely to vary depending on the epilepsy syndrome and seizure types. For example, the intervention threshold may be lower for patients who tend to experience convulsive seizures than for those who experience absence seizures.

Women's choice of thresholds may depend on the additional time and resources required (for example, long-distance travel to access tertiary care) and the perceived risks to themselves and their babies from the various interventions. If the ability to drive is crucial to the mother for her job and other responsibilities, after discussion with clinicians, she may opt for a lower threshold for interventions in secondary care. But if minimising the risk of long-term adverse offspring neurodevelopmental outcomes is valued more by the mother than minimising the risk of seizures, she may choose a higher threshold for increasing the dose and number of anti-epileptic drugs. Our decision curve analysis shows that the model is useful across a wide range of threshold probabilities.

Use of the model in clinical practice should be complementary to individualised advice on safety, risk assessment, drug adherence, and triggers for seizures. Awareness of seizure risk can minimise non-adherence to medication in pregnancy, one of the major factors behind seizure deterioration in pregnancy [6,7,13]. Women predicted to have a low risk of seizure by the model should be informed that their risk status is subject to adherence to their antiepileptic medication. The EMPiRE model does not identify women below 12% risk. Women and clinicians should be aware of this limitation if the probability threshold to make decisions on eligibility for home or water birth is below this threshold.

Relevance to research

The effect of the addition of other markers, such as EEG findings or historical MRI brain imaging reports, on the performance of the model needs further evaluation. There is a need for multiple external validations across different settings and populations to fully appreciate the transportability of the model [54]. The impact of using the EMPiRE model in clinical practice needs to be evaluated through cluster-randomised trials, to assess whether it helps improve outcomes such as the seizure-free period or quality of life of these women. While the tool is expected to improve women's knowledge of their risk status for seizures in pregnancy, the effect of the EMPiRE model on women's anxiety levels is not known and needs to be assessed. Further studies are needed to assess the acceptability of the tool to women with epilepsy and to healthcare providers, their preferred thresholds of choice, and the cost utilities of consequences of decisions for various false positive and false negative cases.

Conclusions

The EMPiRE nomogram is a simple 8-item prediction tool to calculate the individualised risk of seizures at antenatal booking in pregnant women with epilepsy on antiepileptic drugs. The estimates can help guide individually tailored choices made by patients and clinicians, which may influence the intensity of monitoring in pregnancy and after delivery, place of care, and antiepileptic drug dose adjustment strategy. The model is not clinically useful for decision-making at very low thresholds.

Supporting information

S1 Appendix. Flow diagram of EMPiRE trial participants contributing to model development and external validation.

(TIF)

S2 Appendix. Result of sensitivity analysis combining all available data. (DOCX)

S1 TRIPOD Checklist. (DOCX)

Acknowledgments

We are grateful to the original team involved with the collection of the EMPiRE trial data and all women who took part in the study.

Author Contributions

Conceptualization: Ngawai Moss, Khalid S. Khan, Shakila Thangaratinam.

Data curation: John Allotey, Borja M. Fernandez-Felix, Khalid S. Khan, Shakila Thangaratinam.

Formal analysis: John Allotey, Borja M. Fernandez-Felix, Javier Zamora.

Funding acquisition: Khalid S. Khan, Shakila Thangaratinam.

Investigation: John Allotey, Javier Zamora, Manny Bagary, Andrew Kelso, Rehan Khan, Ben W. Mol, Alexander M. Pirie, Dougall McCorry, Khalid S. Khan, Shakila Thangaratinam.

Methodology: John Allotey, Borja M. Fernandez-Felix, Javier Zamora, Joris A. M. van der Post, Ben W. Mol, Dougall McCorry, Khalid S. Khan, Shakila Thangaratinam.

Project administration: John Allotey, Shakila Thangaratinam.

Resources: John Allotey, Khalid S. Khan, Shakila Thangaratinam.

Software: Borja M. Fernandez-Felix.

Supervision: Manny Bagary, Andrew Kelso, Rehan Khan, Ben W. Mol, Alexander M. Pirie, Dougall McCorry, Khalid S. Khan, Shakila Thangaratinam.

Validation: John Allotey, Borja M. Fernandez-Felix, Javier Zamora, Shakila Thangaratinam.

- **Visualization:** John Allotey, Borja M. Fernandez-Felix, Javier Zamora, Ngawai Moss, Khalid S. Khan, Shakila Thangaratinam.
- Writing original draft: John Allotey, Javier Zamora, Shakila Thangaratinam.
- Writing review & editing: John Allotey, Borja M. Fernandez-Felix, Javier Zamora, Ngawai Moss, Manny Bagary, Andrew Kelso, Rehan Khan, Joris A. M. van der Post, Ben W. Mol, Alexander M. Pirie, Dougall McCorry, Khalid S. Khan, Shakila Thangaratinam.

References

- 1. Edey S, Moran N, Nashef L. SUDEP and epilepsy-related mortality in pregnancy. Epilepsia. 2014; 55 (7):e72–4. https://doi.org/10.1111/epi.12621 PMID: 24754364
- Knight M, Nair M, Tuffnell D, Kenyon S, Shakespeare J, Brocklehurst P, et al., editors. Saving lives, improving mothers' care—surveillance of maternal deaths in the UK 2012–14 and lessons learned to inform maternity care from the UK and Ireland Confidential Enquiries into Maternal Deaths and Morbidity 2009–14. Oxford: University of Oxford National Perinatal Epidemiology Unit; 2016.
- Cantwell R, Clutton-Brock T, Cooper G, Dawson A, Drife J, Garrod D, et al. Saving mothers' lives: reviewing maternal deaths to make motherhood safer: 2006–2008. The Eighth Report of the Confidential Enquiries into Maternal Deaths in the United Kingdom. BJOG. 2011; 118 (Suppl 1):1–203. https:// doi.org/10.1111/j.1471-0528.2010.02847.x PMID: 21356004
- 4. Fairgrieve SD. Population based, prospective study of the care of women with epilepsy in pregnancy. BMJ. 2000; 321(7262):674–5. https://doi.org/10.1136/bmj.321.7262.674 PMID: 10987772
- Knight M, Kenyon S, Brocklehurst P, Neilson J, Shakespeare J, Kurinczuk JJ, editors. Saving lives, improving mothers' care: lessons learned to inform future maternity care from the UK and Ireland Confidential Enquiries into Maternal Deaths and Morbidity 2009–12. Oxford: University of Oxford National Perinatal Epidemiology Unit; 2014.
- Man SL, Petersen I, Thompson M, Nazareth I. Antiepileptic drugs during pregnancy in primary care: a UK population based study. PLoS ONE. 2012; 7(12):e52339. https://doi.org/10.1371/journal.pone. 0052339 PMID: 23272239
- Nordeng H, Ystrom E, Einarson A. Perception of risk regarding the use of medications and other exposures during pregnancy. Eur J Clin Pharmacol. 2010; 66(2):207–14. <u>https://doi.org/10.1007/s00228-009-0744-2</u> PMID: 19841915
- Charyton C, Elliott JO, Lu B, Moore JL. The impact of social support on health related quality of life in persons with epilepsy. Epilepsy Behav. 2009; 16(4):640–5. <u>https://doi.org/10.1016/j.yebeh.2009.09.</u> 011 PMID: 19854111
- Loring DW, Meador KJ, Lee GP. Determinants of quality of life in epilepsy. Epilepsy Behav. 2004; 5(6):976–80. https://doi.org/10.1016/j.yebeh.2004.08.019 PMID: 15582847
- Smeets VM, van Lierop BA, Vanhoutvin JP, Aldenkamp AP, Nijhuis FJ. Epilepsy and employment: literature review. Epilepsy Behav. 2007; 10(3):354–62. https://doi.org/10.1016/j.yebeh.2007.02.006 PMID: 17369102
- Royal College of Obstetricians and Gynaecologists Epilepsy in pregnancy. Green-top Guideline No. 68. London: Royal College of Obstetricians and Gynaecologists; 2016 [cited 2019 Apr 16]. https://www. rcog.org.uk/globalassets/documents/guidelines/green-top-guidelines/gtg68_epilepsy.pdf.
- Camfield P, Camfield C. Idiopathic generalized epilepsy with generalized tonic-clonic seizures (IGE-GTC): a population-based cohort with >20 year follow up for medical and social outcome. Epilepsy Behav. 2010; 18(1–2):61–3. https://doi.org/10.1016/j.yebeh.2010.02.014 PMID: 20471324
- Edwards AG, Naik G, Ahmed H, Elwyn GJ, Pickles T, Hood K, et al. Personalised risk communication for informed decision making about taking screening tests. Cochrane Database Syst Rev. 2013;(2): CD001865. https://doi.org/10.1002/14651858.CD001865.pub3 PMID: 23450534
- Battino D, Tomson T, Bonizzoni E, Craig J, Lindhout D, Sabers A, et al. Seizure control and treatment changes in pregnancy: observations from the EURAP epilepsy pregnancy registry. Epilepsia. 2013; 54 (9):1621–7. https://doi.org/10.1111/epi.12302 PMID: 23848605

- Sveberg L, Svalheim S, Taubøll E. The impact of seizures on pregnancy and delivery. Seizure. 2015; 28:35–8. https://doi.org/10.1016/j.seizure.2015.02.020 PMID: 25746572
- Thomas SV, Syam U, Devi JS. Predictors of seizures during pregnancy in women with epilepsy. Epilepsia. 2012; 53(5):e85–8. https://doi.org/10.1111/j.1528-1167.2012.03439.x PMID: 22429269
- Thangaratinam S, Marlin N, Newton S, Weckesser A, Bagary M, Greenhill L, et al. AntiEpileptic drug Monitoring in PREgnancy (EMPiRE): a double-blind randomised trial on effectiveness and acceptability of monitoring strategies. Health Technol Assess. 2018; 22(23):1–152. https://doi.org/10.3310/hta22230 PMID: 29737274
- Altman DG, Vergouwe Y, Royston P, Moons KG. Prognosis and prognostic research: validating a prognostic model. BMJ. 2009; 338:b605. https://doi.org/10.1136/bmj.b605 PMID: 19477892
- Moons KG, Royston P, Vergouwe Y, Grobbee DE, Altman DG. Prognosis and prognostic research: what, why, and how? BMJ. 2009; 338:b375. https://doi.org/10.1136/bmj.b375 PMID: 19237405
- Royston P, Moons KG, Altman DG, Vergouwe Y. Prognosis and prognostic research: developing a prognostic model. BMJ. 2009; 338:b604. https://doi.org/10.1136/bmj.b604 PMID: 19336487
- National Institute for Health and Care Excellence. Epilepsies, diagnosis and management. Clinical guideline CG137. London: National Institute for Clinical Excellence; 2018.
- 22. Harden CL, Hopp J, Ting TY, Pennell PB, French JA, Hauser WA, et al. Practice parameter update: management issues for women with epilepsy—focus on pregnancy (an evidence-based review): obstetrical complications and change in seizure frequency: report of the Quality Standards Subcommittee and Therapeutics and Technology Assessment Subcommittee of the American Academy of Neurology and American Epilepsy Society. Neurology. 2009; 73(2):126–32. <u>https://doi.org/10.1212/WNL.</u> 0b013e3181a6b2f8 PMID: 19398682
- Tomson T, Battino D, Bonizzoni E, Craig J, Lindhout D, Sabers A, et al. Dose-dependent risk of malformations with antiepileptic drugs: an analysis of data from the EURAP epilepsy and pregnancy registry. Lancet Neurol. 2011; 10(7):609–17. https://doi.org/10.1016/S1474-4422(11)70107-7 PMID: 21652013
- Wilhelm J, Morris D, Hotham N. Epilepsy and pregnancy—a review of 98 pregnancies. The Aust N Z J Obstet Gynaecol. 1990; 30(4):290–5. <u>https://doi.org/10.1111/j.1479-828X.1990.tb02013.x</u> PMID: 2082882
- 25. Tomson T, Lindbom U, Ekqvist B, Sundqvist A. Epilepsy and pregnancy: a prospective study of seizure control in relation to free and total plasma concentrations of carbamazepine and phenytoin. Epilepsia. 1994; 35(1):122–30. https://doi.org/10.1111/j.1528-1157.1994.tb02921.x PMID: 8112234
- 26. Bardy AH. Incidence of seizures during pregnancy, labor and puerperium in epileptic women: a prospective study. Acta Neurol Scand. 1987; 75(5):356–60. https://doi.org/10.1111/j.1600-0404.1987. tb05459.x PMID: 3618113
- Fisher RS, Cross JH, D'Souza C, French JA, Haut SR, Higurashi N, et al. Instruction manual for the ILAE 2017 operational classification of seizure types. Epilepsia. 2017; 58(4):531–42. <u>https://doi.org/10.1111/epi.13671</u> PMID: 28276064
- Peduzzi P, Concato J, Kemper E, Holford TR, Feinstein AR. A simulation study of the number of events per variable in logistic regression analysis. J Clin Epidemiol. 1996; 49(12):1373–9. https://doi.org/10. 1016/s0895-4356(96)00236-3 PMID: 8970487
- Moons KG, Altman DG, Reitsma JB. Transparent Reporting of a multivariable prediction model for Individual Prognosis or Diagnosis (TRIPOD): explanation and elaboration. Ann Intern Med. 2015; 162(1): W1–73. https://doi.org/10.7326/M14-0698 PMID: 25560730
- Hosmer DW, Lemeshow S. Applied logistic regression. 2nd edition. New York: John Wiley & Sons; 2000.
- 31. Little RJA, Rubin DB. Statistical analysis with missing data. 2nd edition. New York: John Wiley; 2002.
- **32.** Royston P, Sauerbrei W. Building multivariable regression models with continuous covariates in clinical epidemiology—with an emphasis on fractional polynomials. Method Inform Med. 2005; 44(4):561–71.
- Vergouwe Y, Steyerberg EW, Eijkemans MJ, Habbema JD. Substantial effective sample sizes were required for external validation studies of predictive logistic regression models. J Clin Epidemiol. 2005; 58(5):475–83. https://doi.org/10.1016/j.jclinepi.2004.06.017 PMID: 15845334
- Harrell FE, Lee KL, Mark DB. Multivariable prognostic models: issues in developing models, evaluating assumptions and adequacy, and measuring and reducing errors. Stat Med. 1996; 15(4):361–87. https:// doi.org/10.1002/(SICI)1097-0258(19960229)15:4<361::AID-SIM168>3.0.CO;2-4 PMID: 8668867
- Steyerberg EW, Vickers AJ, Cook NR, Gerds T, Gonen M, Obuchowski N, et al. Assessing the performance of prediction models: a framework for traditional and novel measures. Epidemiology. 2010; 21 (1):128–38. https://doi.org/10.1097/EDE.0b013e3181c30fb2 PMID: 20010215
- Vickers AJ, Elkin EB. Decision curve analysis: a novel method for evaluating prediction models. Med Decis Making. 2006; 26(6):565–74. https://doi.org/10.1177/0272989X06295361 PMID: 17099194

- 37. StataCorp. Stata statistical software. Release 15. College Station (TX): StataCorp; 2015.
- Friedman J, Hastie T, Tibshirani R. Regularization paths for generalized linear models via coordinate descent. J Stat Softw. 2010; 33(1):1–22. PMID: 20808728
- 39. Rubin DB. Multiple imputation for nonresponse in surveys. New York: Wiley; 1987.
- White IR, Royston P, Wood AM. Multiple imputation using chained equations: issues and guidance for practice. Stat Med. 2011; 30(4):377–99. https://doi.org/10.1002/sim.4067 PMID: 21225900
- Aguglia U, Barboni G, Battino D, Cavazzuti GB, Citernesi A, Corosu R, et al. Italian consensus conference on epilepsy and pregnancy, labor and puerperium. Epilepsia. 2009; 50(Suppl 1):7–23. https://doi.org/10.1111/j.1528-1167.2008.01964.x PMID: 19125842
- 42. Thangaratinam S, Rikunenko R, Greenhill L, Bagary M, Pirie A, Khan KS, et al. Optimal monitoring of anti epileptic drugs in pregnancy: time for a randomised controlled trial? Arch Dis Child Fetal Neonatal Ed. 2011; 96(Suppl 1):Fa120. https://doi.org/10.1136/adc.2011.300163.79
- Scottish Intercollegiate Guidelines Network. Diagnosis and management of epilepsy in adults. SIGN 143. Edinburgh: Scottish Intercollegiate Guidelines Network; 2018.
- 44. Harden CL, Pennell PB, Koppel BS, Hovinga CA, Gidal B, Meador KJ, et al. Practice parameter update: management issues for women with epilepsy—focus on pregnancy (an evidence-based review): vitamin K, folic acid, blood levels, and breastfeeding: report of the Quality Standards Subcommittee and Therapeutics and Technology Assessment Subcommittee of the American Academy of Neurology and American Epilepsy Society. Neurology. 2009; 73(2):142–9. https://doi.org/10.1212/WNL. 0b013e3181a6b325 PMID: 19398680
- 45. Moons KG, Kengne AP, Woodward M, Royston P, Vergouwe Y, Altman DG, et al. Risk prediction models: I. Development, internal validation, and assessing the incremental value of a new (bio)marker. Heart. 2012; 98(9):683–90. https://doi.org/10.1136/heartjnl-2011-301246 PMID: 22397945
- 46. Bromley R, Weston J, Adab N, Greenhalgh J, Sanniti A, McKay AJ, et al. Treatment for epilepsy in pregnancy: neurodevelopmental outcomes in the child. Cochrane Database Syst Rev. 2014;(10): CD010236. https://doi.org/10.1002/14651858.CD010236.pub2 PMID: 25354543
- World Health Organization. mhGAP: scaling up care for mental, neurological and substance use disorders. Geneva: World Health Organization; 2011 [cited 2019 Apr 16]. <u>http://www.who.int/mental_</u> health/mhgap_final_english.pdf.
- Harden CL, Hopp J, Ting TY, Pennell PB, French JA, Allen Hauser W, et al. Management issues for women with epilepsy—focus on pregnancy (an evidence-based review): I. Obstetrical complications and change in seizure frequency: report of the Quality Standards Subcommittee and Therapeutics and Technology Assessment Subcommittee of the American Academy of Neurology and the American Epilepsy Society. Epilepsia. 2009; 50(5):1229–36. <u>https://doi.org/10.1111/j.1528-1167.2009.02128.x</u> PMID: 19496807
- 49. Vajda FJ, Hitchcock A, Graham J, O'Brien T, Lander C, Eadie M. Seizure control in antiepileptic drugtreated pregnancy. Epilepsia. 2008; 49(1):172–6. <u>https://doi.org/10.1111/j.1528-1167.2007.01412.x</u> PMID: 18031551
- Collins GS, Ogundimu EO, Altman DG. Sample size considerations for the external validation of a multivariable prognostic model: a resampling study. Stat Med. 2016; 35(2):214–26. https://doi.org/10.1002/ sim.6787 PMID: 26553135
- Lamberink HJ, Otte WM, Geerts AT, Pavlovic M, Ramos-Lizana J, Marson AG, et al. Individualised prediction model of seizure recurrence and long-term outcomes after withdrawal of antiepileptic drugs in seizure-free patients: a systematic review and individual participant data meta-analysis. Lancet Neurol. 2017; 16(7):523–31. https://doi.org/10.1016/S1474-4422(17)30114-X PMID: 28483337
- 52. Kim LG, Johnson TL, Marson AG, Chadwick DW. Prediction of risk of seizure recurrence after a single seizure and early epilepsy: further results from the MESS trial. Lancet Neurol. 2006; 5(4):317–22. https://doi.org/10.1016/S1474-4422(06)70383-0 PMID: 16545748
- Epilepsy Action. A critical time for epilepsy in England: a study of epilepsy service provision in England by Epilepsy Action. Yeadon (UK): Epilepsy Action; 2013 [cited 2019 Apr 16]. https://www.epilepsy.org. uk/sites/epilepsy/files/campaigns/ACT/Epilepsy%20Action%20-%20A%20Critical%20Time%20 (2013).pdf.
- Debray TP, Vergouwe Y, Koffijberg H, Nieboer D, Steyerberg EW, Moons KG. A new framework to enhance the interpretation of external validation studies of clinical prediction models. J Clin Epidemiol. 2015; 68(3):279–89. https://doi.org/10.1016/j.jclinepi.2014.06.018 PMID: 25179855